Author's response to reviews

Title: Mazabraud's Syndrome: First case of fibrous dysplasia of bone associated with soft-tissue myxomas as well as an intra-osseous myxoma: a case report.

Authors:

Wybren M.D. van der Wal (w.vanderwal@maartenskliniek.nl)
Halil M.D. Unal (h.unal@zrt.nl)
Jacky M.D. de Rooy (j.derooy@rad.umcn.nl)
Uta M.D. Flucke (u.flucke@pathol.umcn.nl)
Rene M.D. Ph.D. Veth (r.veth@orthop.umcn.nl)

Version: 4 Date: 24 January 2011

Author's response to reviews:

We hereby submit the revised version of our case report about a patient with Mazabraud's Syndrome.

The reference of the last review is added in the conclusion section.

The numbers of the figures are correct, the names of the files we uploaded correspond to the references in the text. In total 9 figures were uploaded (1a, 1b, 2, 3, 4a, 4b, 5a, 5b, 6).

We hope the article can be published soon now.

Thanks!